Author's response to reviews

Title: Health related quality of life of Dutch children: psychometric properties of the PedsQL in the Netherlands

Authors:

Vivian Engelen (v.engelen@amc.uva.nl)
Marleen M Haentjens (m.m.haentjens@uva.nl)
Symone B Detmar (symone.detmar@tno.nl)
Hendrik M Koopman (h.m.koopman@lumc.nl)
Martha A Grootenhuis (m.a.grootenhuis@amc.uva.nl)

Version: 4 Date: 21 October 2009

Author's response to reviews: see over
Revisions regarding:

**MS: 6162252122466404** "Health related quality of life of Dutch children: psychometric properties of the PedsQL in the Netherlands" Vivian Engelen, Marleen M Haentjens, Symone B Detmar, Hendrik M Koopman and Martha A Grootenhuis

To the editor:

Dear editors,

Thank you very much for accepting our manuscript for publication in BMC Pediatrics (in principal). I have addressed all major and minor revisions, requested in the email of October 15:

- Reference 6, 19 incomplete - Please provide volume and page numbers or article number: done.
- Reference 7 - please provide full details for this reference. For a book the publisher/city/year: done.
- References - The reference list should contain all authors' names, if there are 10 or below. The term 'et al.' should not be used: done.
- List of Abbreviations - Please remove this section as you have no abbreviations: removed.
- Authors' information - Please remove this section as you have no further information: removed.
- Tables - Please move these to a position after the references: done.

Yours sincerely,

Vivian Engelen

Emma Children's Hospital
Academic Medical Center
Psychosocial Department, G8-224
Meibergdreef 9
1105 AZ Amsterdam